\documentclass[10pt,letterpaper, final]{report} % \documentclass{} required in all files

\usepackage{../preamble/myWEPS} % This is now our single preamble file inclusion (it assumes a “.sty” file extension)

\begin{document}

The Configuration Panel window is accessible from the main WEPS screen menu as \textbf{Edit Configuration},
which is the last option under the \textbf{Tools menu}.
It can also be accessed with the \textbf{Alt-C} key combination when the main WEPS screen has focus. 
The Configuration Panel consists of several labeled tabs and three buttons at the bottom of the screen.

The WEPS Configuration Panel buttons are:

\begin{itemize}

\item[]{\textbf{OK} - Saves the changes permanently to the WEPS local configuration file and closes the Configuration Panel window.}

\item[]{\textbf{Cancel} - Cancels all changes and closes the WEPS Configuration Panel window.}

\item[]{\textbf{Help} - Opens the general help for the WEPS Configuration Panel window.}

\end{itemize}

The WEPS Configuration Panel consists of the following tabs:

\begin{itemize}

\item[]{\textbf{Display} -
Provides numerous general settings and options for the WEPS interface.}

\item[]{\textbf{Server} -
Provides settings and options for the server (CSIP cloud) services
which are divided up into three sub-tabs within this panel frame labeled:
\textbf{Model executables}, \textbf{Model options} and \textbf{Remote data services}.}

\item[]{\textbf{Run} -
Specifies the WEPS Run length settings and other related parameters.}

\item[]{\textbf{Soil} -
Specifies the location of the soil database directory, organic soil file
and other options and settings specific to how WEPS treats the soil data provided.}

\item[]{\textbf{Project \& Data Loc} -
A list of directories and configuration files used by the WEPS interface.}

\item[]{\textbf{MCREW} -
Displays the Management/Crop Rotation Editor for WEPS (MCREW) related settings.}

\item[]{\textbf{Windgen} -
Displays the settings pertaining to the Windgen generator, interpolation functions available
and settings on how the interface determines which Windgen record to use for a specified WEPS Run.}

\item[]{\textbf{Cligen} -
Displays the settings pertaining to the Cligen generator and settings
on how the interface determines which Cligen record to use for a specified WEPS Run.}

\item[]{\textbf{WEPS} -
Displays the command-line options for the WEPS science model.}

\item[]{\textbf{GIS} -
Provides the location of the GIS files needed by the MAP tool in the WEPS interface.}

\item[]{\textbf{Reporting} -
Provides setting options for specific detail and debug reports available from the WEPS science model.}

\item[]{\textbf{Email/Mantis} -
Provides settings for the built-in email function and the Mantis ticket system connectivity for the WEPS interface.}

\item[]{\textbf{Fuels} -
Provides access to the Fuels database and displays the current contents of that file.}

\item[]{\textbf{Barriers} -
Provides access to the wind barrier records file path/name.}

\end{itemize}

\section{Configuration Panel - Display Tab} \label{sec:Configuration Panel - Display Tab}

The Display tab is shown here (\cref{fig:WEPS_Configuration_Panel_displaying_the_Display_tab}). It consists of general configuration options for the WEPS interface.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_display_2.jpg}
\caption{WEPS Configuration Panel displaying the Display tab.}
\label{fig:WEPS_Configuration_Panel_displaying_the_Display_tab}
\end{figure}

\begin{figure}[H]
\centering
	\includegraphics[scale=0.7]{images/Configuration/config_display_1.jpg}
\caption{WEPS Configuration Panel displaying the Display tab (old).}
\end{figure}

\begin{itemize}

\item[]{\textbf{Measurement Units} - Display either Metric (SI) or English units on WEPS screens.}

\item[]{\textbf{Time steps per day for wind speed distribution} - 
Enter the number of time steps used for the daily distribution of simulated wind speed.
It cannot be changed and should be greyed out or not displayed at all.} 

\item[]{\textbf{Display latitude and longitude fields} -
Check box to display the latitude and longitude fields in the \textbf{Location Information} panel on the Main screen.
Un-check the box to hide these fields.}

\item[]{\textbf{Display state and county fields} -
Check box to display the state and county fields in the \textbf{Location Information} panel on the Main screen.
Un-check the box to hide these fields.}

\item[]{\textbf{Display elevation} -
Check box to display the elevation field in the \textbf{Location Information} panel on the Main screen.
Un-check the box to hide this field.}

\item[]{\textbf{Display use map button} -
Check box to display the \textbf{Use Map} button in the \textbf{Location Information} panel on the Main screen.
Un-check the box to hide this button.}

\item[]{\textbf{Region slope is read only} -
Check box to only display the simulation region slope field on the main WEPS interface screen and not allow it to be edited.}

\item[]{\textbf{Rock fragments is read only} -
Check box to only display the rock fragments field on the main WEPS interface screen and not allow it to be edited.}

\item[]{\textbf{Barriers are read only} -
Check box to only display the barriers information and not allow them to be edited.}

\item[]{\textbf{Enable full MCREW editing functionality in WEPS} -
Check box to enable the full MCREW editing functionality in WEPS. Allows additional editing functionality of MCREW.
Un-check the box to disable this functionality.
When this functionality is disabled, the MCREW File Open and Open Copy of Template menu items are disabled.}

\item[]{\textbf{Allow estimations in the Soil UI} -
Check box to enable the estimation of soil parameters in the WEPS Soil user interface (UI)
when they are missing from the soil database.}

\item[]{\textbf{Do not display system locale warning at startup} -
Check box to turn off the display of a warning when WEPS is installed in a non-US locale.
To avoid problems, WEPS is always installed using the "US locale", regardless of the local PC settings.
The following is an example of the warning message displayed if the box is not checked and WEPS is installed in a non-US locale:}

\begin{itemize}

\item[]{\textit{WEPS has detected that this machine is using the German (Germany) locale.
WEPS uses the English (United States) locale. You do not need to change your machine's locale,
but be aware that numbers will be formatted in the English (United States) style.
For more info about what a "locale" is, see : www.microsoft.com/globaldev/DrIntl/faqs/Locales.mspx}}

\end{itemize}

\item[]{\texthl{\textbf{Do not estimate missing soil values from SSURGO database} -
Check box to disable the estimation of soil parameters in the user interface when they are missing from the soil database.}}

\item[]{\texthl{\textbf{Organic matter fraction minimum} -
Enter the minimum fraction of organic matter in the soil surface.}}

\item[]{\textbf{Do not prompt to review warnings when restoring a run} -
Check box to turn off warnings displayed from a previous WEPS Run when it is restored.}

\item[]{\textbf{Hide project file buttons} -
Check box to turn off the Project {\scalerel*{\includegraphics{images/icons/folderp.png}}{\strut} button displayed
on the Man and Soil fields on the main WEPS screen.}}

\item[]{\textbf{Hide template file buttons} -
Check box to turn off the Template {\scalerel*{\includegraphics{images/icons/foldert.png}}{\strut} button displayed
on the Man and Soil fields on the main WEPS screen.}}

\item[]{\textbf{Show Countries in Site Chooser} -
Check box to turn on the \textbf{Country} field above the \textbf{State} and 
\textbf{County} fields on the main WEPS screen.}

\item[]{\textbf{Enable experimental XML formats} -
Check box to turn on use of XML formatted files.
This field should not be checked, if displayed for normal WEPS users.}

\item[]{\textbf{Tooltip Delay} -
Sets the delay time for the initial appearance of the tooltip and for the dismissal 
of the tooltip box from the screen.
The units are in milliseconds. To disable tooltip display, set 
the \textit{Initial} value to 1000 (1 second) and the \textit{Dismiss} value to zero.}

\item[]{\textbf{Decimal Precision Passed to Windgen} -
Sets the number of digits right of the decimal to be passed for writing out the lat/lon values
in the weps.run file and for use in the interpolation functions.
This value should be set to 5 for WEPS releases 1.5 or greater.
A value of 2 should be used if strict compatibility with WEPS version 1.3.9 is required.}

\item[]{\textbf{Operation Date} -
Click the down arrow to the right of the box displays a list of available 
formats for the operation date.
The format is displayed on this tab, next to “Operation Date” when a format is selected.}

\item[]{\texthl{\textbf{Search Radius} -
Enter the search radius for the climate station choice lists
(kilometers or miles, depending upon the “Measurement Units” setting).
The user usually will want the nearest station to their simulation site.
The user may want to select a different station more typical of the climate 
for the field being simulated if the nearest station doesn’t meet their criteria.
An example of not selecting the closest station might occur in mountainous areas 
where the nearest station does not always typify the climate for the simulated field.}} 

\end{itemize}

\section{Configuration Panel – Server Tab} \label{sec:Configuration Panel – Server Tab}

\texthl{Server Tab} needs filled in here.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_server_1.jpg}
\caption{WEPS Configuration Panel displaying the Server tab (Model executables sub-tab).}
\label{fig:WEPS_Configuration_Panel_displaying_Server_tab1}
\end{figure}


\section{Configuration Panel – Run tab} \label{sec:Configuration Panel – Run tab}

The \textbf{Run} tab (\cref{fig:Configuration_Panel_displaying_the_Run_tab}) allows the user to specify default required WEPS runtime configuration 
settings.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_run_1.jpg}
\caption{Configuration Panel displaying the Run tab.}
\label{fig:Configuration_Panel_displaying_the_Run_tab}
\end{figure}

\begin{itemize}

\item[]{\textbf{Default Run Mode} options are:}

\begin{itemize}

\item[]{\textbf{NRCS} -
This mode is specified for NRCS users. Note that if this mode is selected,
the user cannot change the run mode from the WEPS interface screen
nor change the number of cycles that are simulated in a WEPS Run.}

\item[]{\textbf{Dates} -
This mode allows the user to set the beginning and ending dates for a simulation run.
It is intended for advanced users and developers who may be wanting
to compare real world field results with what WEPS is simulating.
Thus, it is generally used with real weather data rather than generated data.
Note that the beginning date should be set to January 1 of the calendar year selected
and the ending date being December 31 of the calendar year selected.
Also, the user must be aware that the WEPS interface reports
assume that the length of a management cycle divides evenly into the total simulation runtime length in years.
If not, then those reports will likely be in error.}

\item[]{\textbf{Cycle} -
The mode that ensures that the total WEPS simulation length gets set to a value
that is evenly divisible by the length of the management rotation (rotation years).
The number of “cycles” can be specified by the user on the main WEPS interface screen.
Note that the NRCS mode is identical to \textbf{Cycle} mode,
except the number of cycles is fixed and not allowed to be changed.}

\item[]{\textbf{NRCS Run Length} -
The length of a WEPS simulation run in management rotation “cycles” for an NRCS mode WEPS run.}

\item[]{\textbf{Delete files after run} -
If checked, delete the files listed in the \textbf{File Filter} field at the conclusion of a WEPS run.}

\item[]{\textbf{File Filter} -
The list of files that are to be deleted if the \textbf{Delete files after run} option is enabled.
Display batch script creation option – Determines if the batch script option is presented at the start of a WEPS run.}

\item[]{\textbf{Submodel Outputs} -
Specifies whether the submodel output options are presented to the user at the start of a WEPS run.
The options are: Read Write, Read Only and Not Visible.}

\end{itemize}
\end{itemize}

\section{Configuration Panel – Soil Tab} \label{sec:Configuration Panel – Soil Tab}

The \textbf{Soil} tab allows the user to select the directories
and specify specific options related to WEPS soil data (\cref{fig:WEPS_Configuration_Panel_displaying_the_Soil_tab}.
One can click on the folder icon {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut}
to display a file chooser to select the files or locations for those type of configuration fields.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_soil_1.jpg}
\caption{WEPS Configuration Panel displaying the Soil tab.}
\label{fig:WEPS_Configuration_Panel_displaying_the_Soil_tab}
\end{figure}

\begin{itemize}

\item[]{\textbf{Soil Database} -
The default directory for local “ifc” and NRCS SSURGO Microsoft Access “mdb” soil database files.
It is also where the “SDMDataAccess.nrc.usda.gov.db” file resides containing the link information
to the NRCS online SSURGO soil database.}

\item[]{\textbf{Organic Soil File} -
The default directory location and filename of the WEPS soil file used as a surrogate for organic soils by NRCS.}

\item[]{\textbf{Test for Organic Soils} -
Enables the check for organic soils that is used by NRCS.} 

\item[]{\textbf{Organic Matter Threshold} -
Sets the threshold value for organic matter that the NRCS Organic Soils test uses.}

\item[]{\textbf{Ignore Organic Surface Layer} -
Enables the feature to ignore a surface layer that is high in organic matter when reading in a soil file within WEPS.}

\item[]{\textbf{Maximum Depth of Organic Surface Layer to Ignore} -
Sets the maximum depth for the \textbf {Ignore Organic Surface Layer} feature.}

\item[]{\textbf{Average Stratified Soil Layers} -
Enables the feature to average “stratified” soil layers specified in an NRCS soil data record.}

\item[]{\textbf{Allow Estimations in the Soil User Interface} -
Enables the manually triggered soil property estimation functions in the Soil User Interface
to work if specific soil properties are missing in the displayed soil record.}

\item[]{\textbf{Do Not Estimate Missing SSURGO Database Soil Values} -
Disables the automated soil property estimation functions in WEPS if specific soil properties
are missing in the selected soil record.}

\item[]{\textbf{Soil Editor Readonly} -
Configures the Soil User Interface editor to not allow editing.}

\end{itemize}

\section{Configuration Panel – Project \& Data Loc Tab} \label{sec:Configuration Panel – Directories Tab}

The \textbf{Project \& Data Loc} tab allows the user to select the directories used for templates, skeleton files, databases, and Projects as well as other WEPS related configuration files, etc. by entering the path/file name or clicking the folder icon {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut}  to display a file chooser to select the files or locations (\cref{fig:WEPS_Configuration_Panel_displaying_the_Directories_tab}).

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_directories_1.jpg}
\caption{WEPS Configuration Panel displaying the Directories tab.}
\label{fig:WEPS_Configuration_Panel_displaying_the_Directories_tab}
\end{figure}

\begin{itemize}

\item[]{\textbf{Projects Directory} -
The default directory for WEPS Projects. \$\{user.documents\} is an operating system variable
that defines the user’s documents directory (typically the \textbf{My Documents} folder in Microsoft Windows).}

\item[]{\textbf{New Project Default Runs Locat…} -
The default WEPS Runs location that is used when a new Project directory is specified from within
the WEPS main screen \textbf{Projects} menu.}

\item[]{\textbf{Man Template Open} -
The default directory location for storing management template files.}

\item[]{\textbf{Man Template Save} -
The default directory for saving management template files.}

\item[]{\textbf{Man Conversion Dir} -
The directory specified for use when converting management files 
between the intermediate RUSLE2/WEPS “skel” format and the WEPS format.}

\item[]{\textbf{Operation Records} -
The default directory location for the management operation database files.}

\item[]{\textbf{Crop Records} -
The default directory location for the management crop database files.}

\item[]{\textbf{Field Barriers} -
The default directory location and filename for the WEPS wind barrier data file.}

\item[]{\textbf{MCREW Directory} -
The default directory location for MCREW configuration files.}

\item[]{\textbf{MCREW Data Config} -
The default directory and filename for the MCREW data configuration
file.}

\item[]{\textbf{Management Translation File} -
The translation file used for converting NRCS RUSLE2 “skel" 
format management files to WEPS format management files.}

\item[]{\textbf{Report Image Filename} -
The file name of the image to be displayed on the WEPS report files 
such as an Agency logo.}

\item[]{\textbf{Detail Report Filters} -
The default directory location and filename for the WEPS Tabular Detail Report filters.}

\end{itemize}


\section{Configuration Panel – MCREW Tab} \label{sec:Configuration Panel – MCREW Tab}

\texthl{MCREW Tab} needs filled in here.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_MCREW_1.jpg}
\caption{WEPS Configuration Panel displaying the MCREW tab.}
\label{fig:WEPS_Configuration_Panel_displaying_the_MCREW_tab}
\end{figure}

\section{Configuration Panel – Windgen Tab} \label{sec:Configuration Panel – Windgen Tab}

Displays the settings pertaining to the Windgen generator, interpolation functions available
and settings on how the interface determines which Windgen record to use for a specified WEPS Run
(\cref{fig:WEPS_Configuration_Panel_displaying_Windgen_tab}).

\begin{figure}[H]
\centering
	\includegraphics[scale=0.6]{images/Configuration/config_windgen_1.jpg}
\caption{WEPS Configuration Panel displaying Windgen tab.}
\label{fig:WEPS_Configuration_Panel_displaying_Windgen_tab}
\end{figure}

\begin{itemize}

\item[]{\textbf{Executable} -
Displays the path and file name of the Windgen generator executable.
The folder {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut} button
allows the user to browse and select an alternate Windgen generator executable if desired.}}

\item[]{\textbf{Arguments} -
Displays the Windgen generator executable default command line arguments (see Appendix for argument list).
The “\$\{windgen.data\}” argument option refers to the default location of the Windgen station database file,
which is defined in the \textbf{Database} field on this form.}

\item[]{\textbf{Database} -
Displays the path and file to the Windgen station database file.
The folder {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut} button
allows the user to browse and select an alternate Windgen database file if desired.
The “\$\{weps.databases\}” property is set to the default location where the WEPS database file resides.}}

\item[]{\textbf{Index} -
Displays the path and file to the Windgen station database index file.
This file contains a list of the Windgen stations in the corresponding database file to use in WEPS.
The folder {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut} button allows the user to browse
and select an alternate Windgen database index file if desired.
This file must be kept in sync with the Windgen database file listed above,
otherwise WEPS may not find the desired Windgen station for WEPS Runs.}}

\item[]{\textbf{Interpolate 1} -
Displays the path and file to the spatial interpolation executable file
which determines which three Windgen stations to use for station parameter interpolation at a given location.
The folder {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut} button
allows the user to browse and select an alternate interpolation executable file if desired.}}


\item[]{\textbf{Interpolate 2} -
Displays the path and file to the interpolation executable file which interpolates the station parameters
for the three Windgen stations obtained from the \textbf{Interpolate 1} program listed above.
The folder {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut} button
allows the user to browse and select an alternate interpolation executable file if desired.}}


\item[]{\textbf{Boundary} -
Displays the path and file to the GIS polygon “interpolation boundary” file which specifies
the spatial polygon boundary that defines the region where Windgen stations are interpolated.
The folder {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut} button
allows the user to browse and select an alternate GIS polygon boundary file if desired.}}


\item[]{\textbf{Radius} -
Displays the radius limit that is used in the Windgen station selection field on the main WEPS screen
when the \textbf{Station Choice List} option is active.
It constrains the number of stations to be displayed in the dropdown choice list.}

\item[]{\textbf{Modes} -
Lists the modes that can be enabled for Windgen station selection on the main WEPS screen.
There are several Windgen station selection methods described here:
\textbf{Alternative Weather Station Selection Methods} that employ these modes, when set. The mode options are:}

\begin{itemize}

\item[]{\textbf{Station Choice List} 
Provides a dropdown list of stations sorted by distance from the specified lat/lon on the main WEPS screen,
with the nearest station the default option.}

\item[]{\textbf{Nearest Station} -
Provides the nearest Windgen station to the specified lat/lon coordinates on the main WEPS screen.}


\item[]{\textbf{File} –
Allows the user to select a file that is in the format of a Windgen generator created
Windgen hourly wind data file for use in WEPS.}


\item[]{\textbf{GIS} -
Allows the user to use the GIS polygon map for selecting Windgen stations.}

\item[]{\textbf{Interpolated} -
Allows Windgen station records to be interpolated if they fall within the polygon boundary specified
in the GIS polygon \textbf{Boundary} file listed above.}

\item[]{\textbf{NRCS} -
A special mode that employs several other modes in a prescribed sequence for determining the Windgen station,
based upon the selected lat/lon coordinates specified in the WEPS main screen.
The sequence is defined in the \textbf{Alternative Weather Station Selection Methods} section.}

\end{itemize}
\end{itemize}

\section{Configuration Panel – Cligen Tab} \label{sec:Configuration Panel – Cligen Tab}

Displays the settings pertaining to the Cligen generator, interpolation functions available
and settings on how the interface determines which Cligen record to use for a specified WEPS Run
(\cref{fig:WEPS_Configuration_Panel_displaying_Cligen_tab})

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_cligen_1.jpg}
\caption{WEPS Configuration Panel displaying Cligen tab.}
\label{fig:WEPS_Configuration_Panel_displaying_Cligen_tab}
\end{figure}

\begin{itemize}

\item[]{\textbf{Executable} -
Displays the path and file name of the Cligen generator executable.
The folder {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut} button
allows the user to browse and select an alternate Cligen generator executable if desired.}}

\item[]{\textbf{Arguments} -
Displays the Cligen generator executable default command line arguments (see Appendix for argument list).
The “\$\{cligen.data\}” argument option refers to the default location of the Cligen station database file,
which is defined in the \textbf{Database} field on this form.}

\item[]{\textbf{Database} -
Displays the path and file to the Cligen station database file.
The folder {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut} button allows the user to browse
and select an alternate Cligen database file if desired.
The “\$\{weps.databases\}” property is set to the default location where the WEPS database file resides.}}


\item[]{\textbf{Index} -
Displays the path and file to the Cligen station database index file.
This file contains a list of the Cligen stations in the corresponding database file to use in WEPS.
The folder {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut} button allows the user
to browse and select an alternate Cligen database index file if desired.
This file must be kept in sync with the Cligen database file listed above,
otherwise WEPS may not find the desired Cligen station for WEPS Runs.}}

\item[]{\textbf{Radius} -
Displays the radius limit that is used in the Cligen station selection field on the main WEPS screen
when the Station Choice List option is active.
It constrains the number of stations to be displayed in the drop-down choice list.}

\item[]{\textbf{Modes} -
Lists the modes that can be enabled for Cligen station selection on the main WEPS screen.
There are several Cligen station selection methods described here:
\textbf{Alternative Weather Station Selection Methods} that employ these modes, when set. The mode options are:}

\begin{itemize}

\item[]{\textbf{Station Choice List} -
Provides a dropdown list of stations sorted by distance
from the specified lat/lon on the main WEPS screen, with the nearest station the default option.}

\item[]{\textbf{Nearest Station} -
Provides the nearest Cligen station to the specified lat/long coordinates on the main WEPS screen.}

\item[]{\textbf{File} -
Allows the user to select a file that is in the format of a Cligen generator created Cligen daily weather data file
for use in WEPS.}

\item[]{\textbf{GIS} –
Allows the user to use the GIS polygon map for selecting Cligen stations.}

\item[]{\textbf{NRCS} –
A special mode that employs several other modes in a prescribed sequence for determining the Cligen station,
based upon the selected lat/lon coordinates specified in the WEPS main screen.
The sequence is defined in the \textbf{Alternative Weather Station Selection Methods} section.}

\end{itemize}
\end{itemize}

\section{Configuration Panel - WEPS Tab} \label{sec:Configuration Panel - WEPS Tab}

The WEPS tab is shown here (\cref{fig:WEPS_Configuration_Panel_displaying_the_WEPS_tab}).
It displays the command line options for the WEPS science model and its directory location, etc.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/WEPS_logo.jpg}
\caption{WEPS Configuration Panel displaying the WEPS tab.}
\label{fig:WEPS_Configuration_Panel_displaying_the_WEPS_tab}
\end{figure}

\begin{itemize} 

\item[]{\textbf{Executable} -
Displays the path and file name of the WEPS science executable.
The folder button allows the user to browse and select an alternate WEPS executable if desired.}

\item[]{\textbf{Arguments} -
Displays the WEPS science model command line arguments(see Appendix for argument list).}

\item[]{\textbf{Calibration} -
Display the WEPS science model command line arguments to be used
for a yield calibration run (see Appendix for argument list).}

\end{itemize}


\section{Configuration Panel – GIS tab} \label{sec:Configuration Panel – GIS Tab}

The \textbf{GIS tab} (\cref{fig:WEPS_Configuration_Panel_displaying_the_GIS_tab_main_subtab}) allows the user to specify the GIS directory containing GIS files used in the MAP component that is accessible from the main WEPS interface screen.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_gis_1.jpg}
\caption{WEPS Configuration Panel displaying the GIS tab (main sub-tab).}
\label{fig:WEPS_Configuration_Panel_displaying_the_GIS_tab_main_subtab}
\end{figure}

(\texthl{Needs to be fleshed out here}):

\begin{itemize} 

\item[]{\textbf{GIS Data} –
Sets the directory location that the GIS files used by the MAP component,
which is accessed from the WEPS Interface screen.
The folder icon {\scalerel*{\includegraphics{images/icons/folder.png}}{\strut}
can be clicked on to display a file chooser to select an alternative location, if desired.}}

\end{itemize}

Here is the next "layers" sub-tab (\texthl{Needs to be fleshed out here}):

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_gis_2.jpg}
\caption{WEPS Configuration Panel displaying the GIS tab (layers sub-tab).}
\label{fig:WEPS_Configuration_Panel_displaying_the_GIS_tab_layers_subtab}
\end{figure}

\section{Configuration Panel – Reporting Tab} \label{sec:Configuration Panel – Reporting Tab}

The \textbf{Reporting} tab (\cref{fig:WEPS_Configuration_Panel_displaying_the_Reporting_tab})
allows the user to specify additional WEPS science model output.
The additional output is specified by setting the values to numbers greater than zero for the selected submodels.
Larger numbers produce reports with more detail, if the submodel has multiple levels of reports.
There are two levels of reports, Detail and Debug.
The specific type of output produced at the different levels are provided here: 

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_reporting_1.jpg}
\caption{WEPS Configuration Panel displaying the Reporting tab.}
\label{fig:WEPS_Configuration_Panel_displaying_the_Reporting_tab}
\end{figure}

\begin{itemize}

\item[]{\textbf{Confidence Interval} - Determines whether the WEPS interface Confidence Interval report is made available to users or not.}

\end{itemize} 

\section{Configuration Panel – Email/Mantis Tab} \label{sec:Configuration Panel – Email Tab}

The \textbf{Email} tab (\cref{fig:Configuration_Panel_displaying_the_Email_tab})
allows the user to specify the necessary information for the built-in WEPS email client
and the Mantis bug/comment reporting system.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_email_1.jpg}
\caption{Configuration Panel displaying the Email tab.}
\label{fig:Configuration_Panel_displaying_the_Email_tab}
\end{figure}

\begin{itemize}


\item[]{\textbf{User Name} –
Contains the user name. It is normally set from the WEPS Wizard that runs the first time
a user executes the WEPS interface and requests their name and email address.}

\item[]{\textbf{Email Settings}}

\begin{itemize}

\item[]{\textbf{Use SMTP server (obsolete option)} -
Selecting this option allows the user to use the SMTP server specified in the fields immediately below this option.}

\item[]{\textbf{Use default mail client} 
Selecting this option allows the user to use their default email client,
such as Thunderbird, for sending email messages from WEPS.
Note that this setting does not work for all clients.}

\item[]{\textbf{Use Outlook client} -
Selecting this option allows the user to use their Microsoft Outlook email client when sending email messages from WEPS.
This is the preferred setting for NRCS users.}

\item[]{\textbf{Server Name} -
The default name of the email server to use.}

\item[]{\textbf{Send From} -
The default name of the individual sending the email message.}

\item[]{\textbf{Send Comments To} -
The default email address for sending email comments regarding WEPS.}

\item[]{\textbf{Send Bug Reports} … -
The default email address for sending email bug reports regarding WEPS.}

\end{itemize}

\item[]{\textbf{Mantis Settings}}

\begin{itemize}

\item[]{\textbf{Your email} -
The user’s email address.}

\item[]{\textbf{Mantis URL} -
The URL internet address for the WEPS Mantis ticket system.}

\item[]{\textbf{Mantis user id} -
The Mantis user id (tickets are submitted anonymously from WEPS
so it is not required to fill out this field).
Note that the user can be identified
and replied to if they complete the \textbf{User Name} and \textbf{Your email} fields above.}

\item[]{\textbf{Mantis Password} -
The Mantis user’s password (since tickets are submitted anonymously, 
it is also not required to fill out this field).}

\item[]{\textbf{Mantis project} -
The default Mantis project that submitted tickets are assigned to.}

\end{itemize}
\end{itemize}



\section{Configuration Panel – Fuels tab} \label{sec:Configuration Panel – Fuels tab}

The \textbf{Fuels tab} (\cref{fig:Configuration_Panel_displaying_the_Fuels_tab}
 allows the user to specify default related information in WEPS.

\begin{figure}[H]
\centering
	\includegraphics[scale=0.7]{images/Configuration/config_fuels_1.jpg}
\caption{ Configuration Panel displaying the Fuels tab.}
\label{fig:Configuration_Panel_displaying_the_Fuels_tab}
\end{figure}

\begin{itemize}

\item[]{\textbf{Fuels}} -
Sets the directory location and name for the fuel data file.
The folder icon can be clicked on to display a file chooser to select an alternative location and file, if desired.

\item[]{\textbf{Default}} –
The default fuel type to use for displaying operation fuel usage in WEPS.

\item[]{\textbf{Fuel Table}} –
The fuel table consists of the following columns: 

\begin{itemize}[noitemsep]

\item[]{\textbf{Column 1} – Fuel type.}
\item[]{\textbf{Column 2} – Description of fuel type.}
\item[]{\textbf{Column 3} – Diesel energy equivalent of fuel type.}
\item[]{\textbf{Column 4} – Btu/gal for the specified fuel type.}
\item[]{\textbf{Column 5} – USD/gal for the specified fuel type.}

\end{itemize}
\end{itemize}

\section{Configuration Panel – Barrier Tab} \label{sec:Configuration Panel – Barrier Tab}

\texthl{Barrier Tab} needs filled in here.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.7]{images/Configuration/config_barrier_1.jpg}
\caption{ Configuration Panel displaying the Barriers tab.}
\label{fig:Configuration_Panel_displaying_the_Barriers_tab}
\end{figure}

\end{document}